Author’s response to reviews

Title: Understanding healthcare practices in superdiverse neighbourhoods and developing the concept of welfare bricolage. Protocol of a cross-national mixed-methods study

Authors:

Jenny Phillimore (j.a.phillimore@bham.ac.uk)
Hannah Bradby (hannah.bradby@soc.uu.se)
Michi Knecht (knecht@uni-bremen.de)
Beatriz Padilla (beatriz.padilla@iscte.pt)
Tilman Brand (brand@bips.uni-bremen.de)
Sin Yi Cheung (CheungSY@cardiff.ac.uk)
Simon Pemberton (s.pemberton@keele.ac.uk)
Hajo Zeeb (zeeb@bips.uni-bremen.de)

Version: 2
Date: 15 May 2015

Author’s response to reviews:

I have removed the ethics and funding statement, as requested